

# **JGR** Biogeosciences

# **RESEARCH ARTICLE**

10.1029/2021JG006698

### **Key Points:**

- Both the rate and duration of N
   addition impact soil mineral nutrient
   contents
- Short-term N addition induced different responses in micronutrients, while longer-term treatments decreased total mineral nutrients
- Plant uptake impacted soil available micronutrients at early stage, while soil pH showed stronger effects at later stages

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

Q. Ning and J. Huang, ningqiushi@ibcas.ac.cn; jhhuang@ibcas.ac.cn

#### Citation:

Niu, G., Wang, Y., Wang, R., Ning, Q., Guan, H., Yang, J., et al. (2022). Intensity and duration of nitrogen addition jointly alter soil nutrient availability in a temperate grassland. *Journal of Geophysical Research: Biogeosciences*, *127*, e2021JG006698. https://doi. org/10.1029/2021JG006698

Received 1 NOV 2021 Accepted 7 FEB 2022

### **Author Contributions:**

**Conceptualization:** Guoxiang Niu, Qiushi Ning, Xiankai Lu, Xingguo Han, Jianhui Huang

Data curation: Guoxiang Niu, Yinliu Wang, Ruzhen Wang, Qiushi Ning, Junjie Yang, Xingguo Han, Jianhui Huang Formal analysis: Guoxiang Niu, Yinliu Wang, Ruzhen Wang, Qiushi Ning, Huiling Guan, Junjie Yang, Xiankai Lu, Xingguo Han, Jianhui Huang Funding acquisition: Xingguo Han,

Jianhui Huang Investigation: Guoxiang Niu, Yinliu

Wang, Ruzhen Wang, Qiushi Ning, Junjie Yang, Xiankai Lu, Xingguo Han, Jianhui Huang

Methodology: Guoxiang Niu, Junjie Yang, Jianhui Huang

© 2022. American Geophysical Union. All Rights Reserved.

# Intensity and Duration of Nitrogen Addition Jointly Alter Soil Nutrient Availability in a Temperate Grassland

Guoxiang Niu<sup>1,2,3</sup>, Yinliu Wang<sup>1,3</sup>, Ruzhen Wang<sup>4</sup>, Qiushi Ning<sup>1,3</sup>, Huiling Guan<sup>2,3</sup>, Junjie Yang<sup>1,3</sup>, Xiankai Lu<sup>2,3</sup>, Xingguo Han<sup>1,3</sup>, and Jianhui Huang<sup>1,3</sup>

<sup>1</sup>State Key Laboratory of Vegetation and Environmental Change, Institute of Botany, Chinese Academy of Sciences, Beijing, China, <sup>2</sup>Key Laboratory of Vegetation Restoration and Management of Degraded Ecosystems, South China Botanical Garden, Chinese Academy of Sciences, Guangzhou, China, <sup>3</sup>College of Resources and Environment, University of Chinese Academy of Sciences, Beijing, China, <sup>4</sup>Erguna Forest-Steppe Ecotone Ecosystem Research Station, Institute of Applied Ecology, Chinese Academy of Sciences, Shenyang, China

**Abstract** Increasing N input can alter soil nutrient availability and influence plant growth. Previous studies focused on N addition effects on N and P availability, while less on other mineral nutrients. Besides, how N addition duration affects nutrient availability has remained unclear. Based on a simulative N deposition experiment in a typical steppe with four N addition levels (0, 2, 10, and 50 g m<sup>-2</sup> yr<sup>-1</sup>) under three N addition duration (2, 5, and 10 years), we determined contents of 10 mineral nutrients in surface soils. In the 0–10 cm soil, short-term N addition (2-year) significantly increased exchangeable Ca (+7.2%) and decreased exchangeable Mg (-22.5%) as compared with the control, while decreased available Fe, Cu, and Zn, but increased Mn remarkably (+80.4%). Medium-term N addition (5-year) significantly raised soil total N and available Fe, Mn and Cu, while decreased total P and exchangeable Ca, Na and Mg. The response patterns of these nutrients were largely similar in the 10–20 cm soil, but were weaker and significant only at high N inputs (50 g m<sup>-2</sup> yr<sup>-1</sup>). Long-term N addition (10-year) significantly decreased contents of total base cations (K, Ca, Na, Mg) and micronutrients (Fe, Mn, Cu, Zn) by an average of 32.1% and 20.4%, respectively, across the two soil depths. Influences of pH and plant growth on micronutrients showed remarkable differences among different duration of N addition. These findings indicate that intensity and duration of N addition jointly alter soil nutrient availability and this should be considered in soil nutrient-cycling modeling.

**Plain Language Summary** Multiple soil mineral nutrients support plant growth and help plants to complete their life cycles, but human activities have perturbed global N cycle, which may cause imbalance in the availability of different soil nutrients in ecosystems. In this report we thoroughly assessed the dynamics of essential mineral nutrients for plant growth under different rates and duration of N addition. Short-term N addition (by 2 years) increased availability of Ca and Mn, but decreased the availability of Mg, Fe, Cu, and Zn. Longer-term N addition (by 5 or 10 years) raised soil total N and the availability of Fe, Mn, and Cu, while decreased soil total P and the availability of Ca, Na, and Mg. Changes in soil pH and plant production after N addition were the main reasons to cause abovementioned changes in the availability of those mineral nutrients. These results revealed the existence of accumulative effects of N addition.

# 1. Introduction

Multiple elements sustain plant growth and help plants to complete their life cycles (Kao et al., 2020; Marschner, 2012; Vitousek et al., 2009). These elements include not only carbon (C), hydrogen and oxygen acquired primarily from air and water, but also include nitrogen (N), phosphorus (P), base cations and micronutrients, which are obtained primarily from mineral soils (Watanabe et al., 2007). All those mineral elements play a unique role in plant health and ecosystem functioning, and are also regarded as limiting factors of plant productivity (Fang et al., 2017; Kao et al., 2020; Waldron et al., 2009). For example, the primary productivity of terrestrial ecosystems is widely limited by either N or P, or by both (Du et al., 2020; Hou et al., 2021). Calcium (Ca) and magnesium (Mg) can not only regulate soil pH especially in alkaline soil, but also play important roles in processes of cell signal transduction and C assimilation (Bowman et al., 2008; X. K. Lu et al., 2018; Verbruggen & Hermans, 2013). Deficiency of some micronutrients, such as iron (Fe) and manganese (Mn), could often induce negative effects on plant enzyme synthesis, but their high concentrations could also cause toxicity for growth of plants and microorganisms (Kao et al., 2020; Tian et al., 2020; Waldron et al., 2009). Currently, high



**Project Administration:** Xingguo Han, Jianhui Huang

Resources: Guoxiang Niu, Yinliu Wang, Ruzhen Wang, Xiankai Lu, Xingguo Han Software: Guoxiang Niu, Ruzhen Wang, Qiushi Ning, Huiling Guan

Supervision: Xingguo Han, Jianhui Huang

Validation: Guoxiang Niu, Huiling Guan Visualization: Guoxiang Niu, Ruzhen Wang, Qiushi Ning, Huiling Guan, Junjie Yang, Xiankai Lu

Writing – original draft: Guoxiang Niu Writing – review & editing: Guoxiang Niu, Ruzhen Wang, Qiushi Ning, Junjie Yang, Xiankai Lu, Xingguo Han, Jianhui Huang reactive N inputs to soils induced by human activities might cause some unpredictable changes in the availability of those mineral nutrients (Galloway et al., 2008; Wen et al., 2020). However, dynamics of the full set of these mineral nutrients under increasing and long-lasting N inputs to date remain largely unknown, especially for those soil base cations and micronutrients.

In N enriched soils, the base cations (K, Ca, Na, and Mg) binding to soil surface can be replaced by the  $NH_4^+$  ions from the N fertilizer, thereby leached out from soils (Cameron et al., 2013). Following the continuous N addition and depletion of base cations, the micronutrients (Fe, Mn, Cu, and Zn) are mobilized for acid buffering, with toxic effects on plant and microorganisms (Bowman et al., 2008; Keiblinger et al., 2018; Lucas et al., 2011). The effects of N inputs on the contents of mineral elements may depend on both intensity and duration of N inputs (Chen et al., 2020; Lucas et al., 2011; Meng et al., 2019). Generally, N addition has increasing effects on contents of soil available N and some micronutrients, and decreasing effects on contents of P and base cations, and all those effects were usually amplified with the rates and duration of N inputs (Deng et al., 2017; Hou et al., 2018; M. Lu et al., 2011; Lucas et al., 2011). However, there is no consistency towards the N effects on the above-mentioned findings. Taking soil P as an example, Chen et al. (2020) reported that N addition had no significant effects on the contents of soil available and total P whether in short- or in long-term studies, while soil phosphatase activity was diminished over time by N addition. Results from a meta-analysis also showed that the decreasing effects of N fertilization on base cations in terrestrial ecosystems only occurred in the time periods of first-5 years (Lucas et al., 2011). Collectively, the above inconsistency may result from that (a) the studies include soils with heterogeneous properties and different vegetation types and (b) the studies have been conducted with one or two N addition rates in short-term. Hence, studies that carried out with soils from the exact same origin and with varied rates and durations of N addition are needed to systematically test the effects of N addition on soil mineral nutrients.

Increased use of fertilizers in agricultural activities, and excessive consumption of fossil fuel in industries have led to high reactive N emission to the atmosphere over the last century, which further results in enhanced anthropogenic N deposition to terrestrial ecosystems (Fowler et al., 2013; Galloway et al., 2008; Wen et al., 2020). Increase in N deposition would lead to a series of consequences, causing changes in plant biomass, microbial activities and mineral elements contents in grassland ecosystems (Bengtsson et al., 2019; Greaver et al., 2016; Vitousek, 2015). Importantly, N addition might also indirectly affect soil mineral nutrients through altering plant properties and soil pH (Kao et al., 2020; Liu et al., 2021; Meng et al., 2019; Tian et al., 2019). For example, Stevens (2015) reported that the aboveground net primary production (ANPP) was positively correlated with dose of atmospheric N deposition in herbaceous ecosystems, and the increase by 1 g N m<sup>-2</sup> yr<sup>-1</sup> could raise ANPP up by 3% globally. Therefore, the plant uptake of mineral nutrients for growth under N enrichment might profoundly influence dynamics of mineral nutrients in soils. Nevertheless, the N addition, especially at high rates, can also lower soil pH which may further affect the contents of base cations and micronutrients (Liu et al., 2021; Meng et al., 2019). The N-addition effects on plant growth and soil pH also vary with different duration time. In a dosage accumulation analysis, Phoenix et al. (2012) demonstrated that even very low N deposition rates can result in significant ecological changes of plant and soil properties if the N loading continued long enough. Thus, the contents of mineral elements are expected to undergo cumulative effects over longer term N addition. However, the regulatory roles of plant and soil pH in changes of soil mineral nutrients under scenarios of increasing N deposition remain unclear. In addition, although ecologists can predict effects of N addition with different intensities and durations on terrestrial ecosystems based on results of meta-analysis, evidence from the long-term field experiments still remains scarce (Clark & Tilman, 2008; Lucas, et al., 2011; Meng et al., 2019).

China's grassland covers 12.5% of the global grassland area, or ~41% of terrestrial China, and they have played important roles in provisions of ecosystem service and functioning (Bengtsson et al., 2019). However, they are also confronted some environmental issues, such as increasing N deposition (Wen et al., 2020). Most previous studies addressed effects of multiple N addition rates on the species composition and biomass production of vegetation (Bai et al., 2010; Clark & Tilman, 2008; Niu et al., 2018), yet the data is still scarce on the soil properties and nutrient availability, especially under different duration of N application. Furthermore, although huge progresses have been made in understanding the dynamics of mineral elements in grassland region, most of them mainly focused on some single mineral element with only snapshot sampling, which might only provide us some fragmentary images (Fang et al., 2017; Horswill et al., 2008; Liu et al., 2021; Lucas et al., 2011). Therefore, evidences from the long-term field experiments across different N addition rates are required for accurately predict the response of primary soil mineral nutrients to N enrichment in grassland. In this study, we assessed



the dynamics of 10 mineral elements (including N, P, base cations K, Ca, Na, and Mg, and micronutrients Fe, Mn, Cu, and Zn) in surface soils (0–10 and 10–20 cm) using a long-term field N addition experiment with four N addition levels (0, 2, 10, and 50 g N m<sup>-2</sup> yr<sup>-1</sup>) under three N addition durations (2, 5, and 10 years). Based on the results of previous studies (Liu et al., 2021; Lucas et al., 2011; 5R. Wang et al., 2018), we hypothesized that, (a) N addition would increase contents of total N and exchangeable base cations, but decrease contents of total P and available micronutrients in this N-deficient grassland; (b) N addition effects would be strengthened and accumulated with increasing rates and prolonged durations of N addition; (c) The effects of soil pH on changes of mineral elements would become stronger with increasing duration of N addition as the progressive soil acidification.

# 2. Material and Methods

### 2.1. Study Site and Experimental Set Up

The field experiment was conducted in a fenced temperate semiarid grassland site of the Inner Mongolia Grassland Ecosystem Research Station (IMGERS; 116°40′E, 43°13′N; elevation 1,200 m). The mean annual temperature was 0.9°C (1985–2017 of IMGERS) and the mean annual precipitation was 321.8 mm with ~70% of the precipitation falling in the growing season from May to August. The soil type is a Haplic Calcisols according to the FAO Soil Taxonomy classification with percent contents of clay, silt and sand in the 0–10 cm soil being 5.2%, 44.2%, and 50.6%, respectively (Niu, Wang, et al., 2021). *Leymus chinensis* and *Stipa grandis* are the two dominant plant species in the vegetation, and they account for over 60% of total community ANPP which generally peaks in late August. The background amount of atmospheric N deposition in the study area is less than 2 g N m<sup>-2</sup> yr<sup>-1</sup> (G. R. Yu et al., 2019).

The N addition experiment was established in September 2008 and N addition treatment has been implemented yearly since then. To exclude disturbance from large livestock (i.e., cattle and sheep) grazing, the experimental site has been fenced since 1999. The whole experiment was set up as a completely randomized block design with 10 blocks as replicates. In each block, the treatment included 9 levels of N addition (0, 1, 2, 3, 5, 10, 15, 20, and 50 g N m<sup>-2</sup> yr<sup>-1</sup>) crossed with two addition frequency (half-yearly and monthly) and two levels of mowing (mown and unmown). The detailed experimental design was described in Niu, Wang, et al. (2021). In this study, only four N addition rates (0, 2, 10, and 50 g N m<sup>-2</sup> yr<sup>-1</sup> with half-yearly addition and unmown treatments, representing control, low-N, mid-N, and high-N addition, respectively) in 6 out of 10 blocks with three durations (2-, 5-, and 10-year, representing short-term, medium-term, and long-term N addition treatment, respectively) were selected to assess the effects of intensity and duration of N addition on abovementioned mineral nutrients. The atmospheric N deposition at the rate of 2 g N m<sup>-2</sup> yr<sup>-1</sup> was close to the background atmospheric N deposition and the 10 g N m<sup>-2</sup> yr<sup>-1</sup> was considered as the critical N saturation threshold of ecosystem functioning in this region (Bai et al., 2010; G. R. Yu et al., 2019). The level of 50 g N m<sup>-2</sup> yr<sup>-1</sup> was also applied to simulate extremely high N fertilization scenario in this agro-pastoral region (Niu, Hasi, et al., 2021). We selected three durations of time based on a previous study with similar vegetation by Tian et al. (2020), who divided the dynamics of ecosystem structure and functioning under N enrichment into three stages, that is, early ( $\sim 1-3$  years), middle ( $\sim 4-9$  years) and late stage ( $\sim$ 10–15 years). The N was added in the form of NH<sub>4</sub>NO<sub>3</sub> equally on the first day of June (in solution by spraying) and November (spread by hand). It should be noted that no N and other nutrients were added prior to our experiment, and we did not add any other nutrients (e.g., K and Fe) except for N in our system, thus limitation by other nutrients may occur over time.

### 2.2. Plant and Soil Sampling

In August of 2010, 2013 and 2018 (after consecutive N addition of 2, 5, and 10 years, respectively), ANPP of each plant species in a  $0.5 \times 2$  m quadrat was harvested by clipping vegetation above the ground, while roots or below-ground biomass (BGB) (by depth of 0–10 cm and 10–20 cm) were sampled by collecting soil samples using core with diameter of 6.8 cm. The composite soil samples by soil depth (0–10 and 10–20 cm) were randomly collected at five locations and mixed in each plot using a core (diameter in 5 cm). To remove rocks and visible roots, the fresh soil samples were passed through a 2 mm sieve, and air-dried for determining contents of soil exchangeable or available mineral nutrients, physicochemical properties, and contents of total soil mineral nutrients.



### 2.3. Laboratory Analyses

All plant materials were oven-dried at 65°C for 48 hr and weighed to calculate ANPP and BGB. Soil pH was measured with a pH meter (sartorius PB-10; Goettingen, Germany) in a 1:2.5 (wt/vol) ratio of soil-to-water extract. The content of total N (TN) was determined with an elemental analyzer (Analytik-jena multi N/C3100; Jena, Germany). An inductively coupled plasma optical emission spectrometer (ICP-OES, Thermo Electron Corporation, USA) was used to measure the contents of total amount of P (TP), potassium (TK), calcium (TCa), sodium T (Na), magnesium (Mg), ferrum (TFe), manganese (TMn), cuprum (TCu) and zinc (TZn) in soils digested with strong acid mixture (HCl + HF + HClO). The exchangeable amount of K, Ca, Na and Mg were determined by extracting soil samples with 1 mol L<sup>-1</sup> ammonium acetate solution in a 1:10 (wt/vol) ratio of soil to solution (Ochoa-Hueso et al., 2014). Briefly, a 5 g air-dried soil sample was extracted with 50 ml ammonium acetate (pH = 7.0) and shaken at 200 rpm for half-hour. The mixture was filtered and the concentrations of aforementioned exchangeable base cations in filtrates were measured by ICP-OES. Soil available Fe, Mn, Cu, and Zn were extracted with a mixture solution (including 0.005 mol L<sup>-1</sup> diethylenetriaminepentadcetic acid, 0.01 mol L<sup>-1</sup> CaCl<sub>2</sub> and 0.1 mol L<sup>-1</sup> triethanolamine) in a 1:3 (wt/vol) ratio of soil to solution (Haddad & Evans, 1993; Lindsay & Norvell, 1978). The mixed soil solution was also shaken and filtered to pass through filter paper, and the concentrations of available micronutrients were also determined by ICP-OES.

### 2.4. Statistical Analyses

All statistical analyses were executed in R4.0.5 (https://www.r-project.org/2021-3-31) and all data were tested whether to meet the requirement of statistical analyses in "Car" package. Then, two-way ANOVA were conducted to assess the effects of the amount and duration of N addition on soil nutrients respectively in 0–10 and 10–20 cm soil layers. One-way ANOVA with Tukey HSD post-hoc test was used to test the effects of different N addition rates or different N addition duration on mineral nutrients. Soil pH controls the transfer processes (hydrolysis and sorption) of soil mineral nutrients, while plant biomass can reflect the amount of plant uptake to these nutrients (Bowman et al., 2008; Cai et al., 2017; Meng et al., 2019). Thus, they both are important factors in regulating the availability of soil nutrients. To determine the relative influence of soil pH and plant biomass (including ANPP and BGB) on the changes of mineral nutrients and indicate the specific pathways of the two variables, we run variation partitioning analysis (VPA) in the "Vegan" package and structure equation model (SEM) in the "piecewiseSEM" package under different N addition duration, respectively (Fang et al., 2019; Liu et al., 2021; B. Wang et al., 2021). Because soil mineral nutrients in the SEM include 10 nutrients, we further show the significant responses ( $P \le 0.05$ ) of specific nutrients by combing the results of ANOVA. Pearson correlation analyses were conducted to explore the potential correlations of mineral nutrients with ANPP, BGB, and pH in both soil layers.

### 3. Results

### 3.1. Soil Total N and P

The duration of N addition significantly affected the contents of TN and TP in both 0–10 and 10–20 cm soil layers. Long-term N addition (10 years) increased TN content by 15.6% and decreased TP content by 16.6% across the two soil layers as compared with those of short-term (2-year) N addition (Figure 1). In the 0–10 cm soil, high-N addition significantly increased TN content by 32.4% after 10-year N addition as compared with the control (Figure 1a), while significant decrease of TP content were observed in all N addition levels especially after 5- and 10-year N addition (Figure 1c). In the 10–20 cm soil, TN content increased along with N addition in all N addition durations and was accumulative with increasing durations (Figure 1b), while high-N addition significantly decreased TP content only after 5- and 10-year N addition (Figure 1).

### 3.2. Soil Base Cations

Both the amount and duration of N addition (either singly or interactively) significantly affected the contents of exchangeable Ca, Na, and Mg in the both soil layers (Figure 2). The high N addition rates had strong negative effects on contents of soil exchangeable Ca with all the N addition durations in the 0–10 cm soil layer, and with 10-year N addition duration in the 10–20 cm soil layer (Figures 2c and 2d). Comparing with the control, mid-N addition significantly increased the exchangeable Ca content after 2-year N addition in both soil layers. In the



# **Journal of Geophysical Research: Biogeosciences**



**Figure 1.** Effects of rates and duration of nitrogen addition on soil total N (TN) and P (TP) contents. Notes: Different capital letters denote significant differences ( $P \le 0.05$ ) among different durations, and different lowercase letters mean significant differences ( $P \le 0.05$ ) among different treatments in a given year. The results of two-way ANOVA were shown at the top of each figure. The letter ns and symbols \*, \*\*, \*\*\* indicate insignificant or significant differences at  $P \le 0.05$ , 0.01, 0.001, respectively (F values, n = 6).

0–10 cm soil layers, high N addition rate with 10-year duration remarkably reduced soil exchangeable Na content, and the combination of longer duration and high rates of N addition tended to result in lower soil exchangeable Na content in the 10–20 cm soil (Figures 2e and 2f). The high-N addition significantly decreased the exchangeable Mg content in the 0–10 cm soil after 2- and 10-year N addition, and in the 10–20 cm soil after 10-year N addition (Figures 2g and 2h).

The 5- and 10-year N addition negatively affected total K, Ca, Na, and Mg in both soil layers, which was opposite with those after 2-year N addition (FigureS1 in Supporting Information S1). In the 0–10 cm soil layers, the total contents of K, Ca, Na, and Mg increased significantly with N addition rates after 2-year N addition, but they decreased significantly after 5- and 10-year N addition (FigureS1 in Supporting Information S1). When comparing with the results of 2-year N addition across the two soil depths, the contents of total K, Ca, Na, and Mg after 10-year N addition significantly decreased by 9.3%, 41.8%, 23.8%, and 53.5%, respectively (Figure S1 in Supporting Information S1).

# 3.3. Soil Micronutrients

The rates and duration of N addition, either singly or interactively, significantly increased the concentrations of available micronutrients in both soil layers (Figure 3). After 2-year N addition, low- and mid-N addition generally decreased the contents of available Fe, Mn, Cu, and Zn in the 0–10 cm soil as compared with the control, but showed an increasing tendency with high-N addition (Figure 3). This is especially true for Mn, the content of





Figure 2. Effects of duration and amounts of nitrogen addition on soil exchangeable K (Ex. K), Ca (Ex. Ca), Na (Ex. Na), and Mg (Ex. Mg) contents in a typical steppe. See other notes in Figure 1.



# Journal of Geophysical Research: Biogeosciences

10.1029/2021JG006698



Figure 3. Effects of duration and amounts of nitrogen addition on the concentrations of soil available Fe (Av. Fe), Mn (Av. Mn), Cu (Av. Cu), and Zn (Av. Zn). See other notes in Figure 1.



which increased remarkably (+285.4%) (Figure 3c). After 5- and 10-year N addition, the contents of available Fe, Mn, and Cu increased strongly with mid- and high-N addition (Figures 3a, 3c, and 3e). Significant increase for Zn was also found with low-N addition (5-year) and high-N addition (10-year) in the 0–10 cm soil (Figure 3g). The pattern was largely similar for the behavior of these micronutrients in the 10–20 cm soil, but the increasing amplitude was smaller with high rates of and under long-term N addition especially for Fe and Mn (Figures 3b and 3d).

The rates and duration of N addition showed inconsistent effects on the total amount of these micronutrients (Figure S2 in Supporting Information S1). The total contents of Fe and Mn declined overall in both soil layers after 5- and 10-year N addition (Figure S2b, d in Supporting Information S1), while both insensitive responses were found for Cu and Zn to the rates of N addition (Figure S2f, h in Supporting Information S1). The 10-year N addition decreased the contents of total Fe, Mn, Cu, and Zn, respectively, by 20.5%, 33.9%, 14.8%, 12.4% overall in comparison with those under 2-year N addition in both soil layers (Figure S2 in Supporting Information S1).

### 3.4. Correlations of Mineral Elements With ANPP, BGB, and Soil pH

Results of VPA showed that the relative contribution of soil pH on total variation of the 10 mineral nutrients in the 0–10 cm soil increased with duration of N addition, and the relative contribution of soil pH was all significant after 5- and 10-year N addition, which accounted for 45.7% and 42.7% of total variation, respectively (Figures 4a–4c). The relative contribution of plant biomass (including both ANPP and BGB) was smaller than soil pH in regulating the response of soil mineral nutrients to 5- and 10-year N addition (Figures 4a–4c). The subsequent SEM analysis also suggested that soil pH was the main pathway affecting soil nutrient availability after 5- and 10-year N addition, and the exchangeable Ca and Mg, and the available Fe, Mn, and Cu were more sensitive to N addition regardless of N addition duration (Figures 4d–4f). Overall, the relative contribution of soil pH was higher than plant biomass (Figure 4d) for the variation of soil mineral nutrients, and the unexplained part decreased over time (Figure 4).

When considering all data of the three sampling years and based on results of Pearson correlation analyses, we further explored the potential correlations of each mineral nutrient with ANPP, BGB, and pH over time. Soil TN was positively correlated with BGB and negatively correlated with pH in the 0–10 cm soil (all  $P \le 0.05$ , Table 1), while significantly negative correlations of TN with ANPP, BGB and pH, and TP with ANPP were detected in the 10–20 cm soil (Table 1). When considering data by sampling year, soil TN was positively correlated with ANPP in the 0–10 cm soil after 2-year N addition and negatively correlated with pH in both 0–10 and 10–20 cm soils (all  $P \le 0.05$ , Table 1). Soil TP correlated positively with BGB in the 0–10 cm soil and negatively with pH in the 10–20 cm soil (all  $P \le 0.05$ , Table 1). After 5-year N addition, soil TN correlated positively with BGB in the 0–10 cm soil and positively with ANPP in the 10–20 cm soil, while significantly negative correlations of pH with TN and positive correlations of pH with TP were also found in both soil layers (Table 1). Soil TP was negatively correlated with ANPP and BGB in the 0–10 cm soils (Table 1). After 10-year N addition, soil pH correlated negatively with TP in both soil layers (all  $P \le 0.05$ , Table 1).

Generally, ANPP was positively correlated with the content of available Mn, and negatively with available Zn in the 0–10 cm soil (all  $P \le 0.05$ , Table 2). Soil pH was also positively correlated with contents of exchangeable Mg, Ca and Na, and negatively with available Fe, Mn and Cu in both soil layers (all  $P \le 0.05$ , Table 2). In the 0–10 cm soil, ANPP was positively correlated with Mn content after 2-year N addition, and soil pH also correlated negatively with Mn content (all  $P \le 0.05$ , Table 2). The ANPP and BGB was positively correlated with available Fe, Mn, and Cu after 5-year N addition, while soil pH was negatively correlated with these micronutrients (all  $P \le 0.05$ , Table 2). The BGB was positively correlated with the contents of available Zn, Mn, and Cu under 10-year N addition, while pH was positively correlated with exchangeable Ca and Mg, and negatively correlated with available Fe, Mn, and Zn in the 0–10 cm soil (Table 2).

### 4. Discussion

### 4.1. Effects of Intensity and Duration of N Addition on Soil TN and TP Contents

As we hypothesized, the intensifying effects of high N addition rate and cumulative effects of long-term N addition were shown for both soil TN (increase) in both soil layers and TP (decrease) in the 0–10 cm soil. It has been well established that the N addition could affect soil N pools by both plant uptake and soil N cycling processes





**Figure 4.** Variation partitioning analysis (VPA, a–c) and structural equation model (SEM) analysis (d–f) of the effects of N addition on soil nutrient availability in 0–10 cm soil under different N addition duration (2-year, 5-year, and 10-year). Results of VPA can indicate the individual relative contribution of soil pH, plant biomass and their intersection on total variation of 10 mineral element, while SEM analysis can further indicate the specific pathways. Negative values in the VPA were assigned as zero. Results of SEM model fitting by using "dsep" and "fisherC" test indicate a good fit of the model to data. The solid and dashed arrows indicate significant ( $P \le 0.1$ ) and insignificant effects, respectively, while  $r^2$  values indicate the proportion of variation explained by relationships with other variables. Values associateds with arrows indicate standardized path coefficients. The symbols " $\downarrow$ " and " $\uparrow$ ," respectively, indicate a significant decrease or increase ( $P \le 0.05$ ) in response to N addition, and the number in each square box indicates the response ratio of soil nutrients to N addition.

(M. Lu et al., 2011; Tian et al., 2020). Previous studies showed that N addition significantly increased the N stocks of plant and litter, but no significant increase was observed in topsoil (M. Lu et al., 2011; Ren et al., 2021). In this study, our results showed that only high-N addition could significantly increase soil TN content in both surface soil layers, with more pronounced responses being observed under long-term N addition (Figure 1). These results suggested that the effects of N addition on the soil N stocks are determined not only by the intensity of additional N input but also by the accumulation of the effects through years. The fact that soil TN contents lacked response to lower rates and relative short-term N addition probably resulted from enhanced plant N uptake in this N-limiting system (Bai et al., 2010; Ren et al., 2021) and gaseous losses from soils via denitrification (Boy-Roura et al., 2015; Cameron et al., 2013; Zhang et al., 2018; Zia et al., 2020). In this study, we also found that high-N addition significantly increased TN content in the 10–20 cm soil in all sampling years, indicating more N leached down especially under high N addition rate. Our previous study also reported that the contents of NO<sub>3</sub><sup>-</sup>-N, NH<sub>4</sub><sup>+</sup>-N and DON in N addition plots were significantly higher as compared with those in control plots even down to the 70–100 cm depth of soil profiles (Niu, Hasi, et al., 2021).

The significant reduction of TP content in the 0–10 cm soil may suggest that P limitation will occur under continuous N enrichment scenario. Our results are consistent with the main findings of previous meta-analysis (Deng et al., 2017; Hou et al., 2018; Li et al., 2016). Two biogeochemical processes linking P cycling may help to explain our results. First, N addition can enhance plant production and consequently stimulate plant P uptake



Pearson Correlations Among Soil Total N (TN) and P (TP) Contents With Above-Ground Net Primary Production (ANPP) Below-Ground Biomass (BGB) and pH

Factors	Duration	TN	ТР	Factors	Duration	TN	TP	
(a) 0–10 cm				(b) 10–20 cm				
ANPP	2-year	0.36	0.35	ANPP	2-year	0.60**	0.33	
	5-year	0.20	-0.44*		5-year	0.50*	-0.44*	
	10-year	0.20	-0.32		10-year	0.35	-0.25	
	Overall	-0.21	0.19		Overall	-0.23*	-0.53***	
BGB	2-year	-0.23	0.44*	BGB	2-year	0.51	-0.09	
	5-year	0.40*	-0.40*		5-year	0.27	0.14	
	10-year	0.31	0.15		10-year	0.05	0.03	
	Overall	0.24*	0.16		Overall	-0.27*	-0.01	
pН	2-year	-0.42*	-0.25	pН	2-year -0.28**		-0.57**	
	5-year	-0.50*	0.60**		5-year	-0.45*	0.64***	
	10-year	-0.87***	0.37		10-year	-0.63***	0.30	
	Overall	-0.60***	0.36**		Overall	-0.66***	-0.01	

*Note.* Symbols \*, \*\*, and \*\*\* mean significance at  $P \le 0.05$ , 0.01 and 0.001. Significant ( $P \le 0.05$ ) data were presented in bold.

(Deng et al., 2017; Marklein & Houlton, 2012), which will potentially increase P limitation in this grassland region. This is also supported by the negative correlations of ANPP with TP especially under longer-term (5- and 10-year) N addition (Table 2), indicating the exhaustion of P for plant growth. Second, N addition can cause soil acidification especially with high-N inputs and under long-term N addition, which further activates some metal ions, such as available Fe and Al (Barrow, 2017; Bowman et al., 2008; Fang et al., 2019; Spohn, 2020). Thus, the binding process of phosphate with Fe and Al ions may be accelerated by soil acidification and thereby contribute to the decreased P availability (Barrow, 2017; Spohn, 2020).

### 4.2. The Intensity and Duration of N Addition Jointly Affect Soil Base Cations

The longer-term (i.e., 5- and 10-year) N addition significantly decreased the contents of exchangeable Ca, Na, and Mg especially at mid- and high-N addition (Figure 2), which partly supported our second hypothesis. However, an unexpected finding was also observed that short-term N addition significantly increased the total contents of base cations and exchangeable Ca (Figure 2). Such finding is mainly because the impacts of external disturbance in this natural ecosystem, such as the excessive N supply in the closed system, may accelerate weathering process and physical fragmentation of rocks (Doetterl et al., 2018; Oliveira Garcia et al., 2020; Rafiei & Kennedy, 2020). Nevertheless, some other processes may also cause changes in the total and exchangeable contents of base cations, including plant uptake, binding and dissolution with soil organic materials, and leaching out from the soil system (Bünemann et al., 2018; Cameron et al., 2013; R. Wang et al., 2018). In our study area, the soil is weakly alkaline with relatively high exchangeable contents of Ca, Na, and Mg (Niu, Wang, et al., 2021) and thus has high capacity to buffer N-induced acidification. Besides, uptake of these cations by plants is relatively low, as indicated by low ANPP under either short-term N addition or low N inputs. Therefore, the sensitive responses of these cations to either short-term N addition or lower N inputs hardly occur. Moreover, with combination of dynamics of soil NO<sub>3</sub><sup>--</sup>N, the amounts of base cations leaching out from upper to deeper soil layers would also increase with increasing rates and duration of N addition (Cameron et al., 2013; Doetterl et al., 2018; Niu, Hasi, et al., 2021). Our previous study in the same region also found that 13-year N addition only decreased the contents of base cations in surface soils while had insignificant effects on those in subsoils (Niu, Hasi, et al., 2021). The significant decrease in soil total and exchangeable contents of base cations under long-term N addition demonstrated that the decrease in supply potential of soil base cations to plants, which may cause nutrient imbalance and impoverishment under chronic N and incremental N enrichment (Fang et al., 2017; Lucas et al., 2011; R. Wang et al., 2018).



### Table 2

Pearson Correlations Among the Contents of Soil Exchangeable K (Ex K), Ca (Ex. Ca), Na (Ex. Na), and Mg (Ex. Mg) and Available Fe (Av. Fe), Mn (Av. Mn), Cu (Av. Cu), and Zn (Av. Zn) With Above-Ground Net Primary Production (ANPP), Belowground Biomass (BGB), and pH

Factors	Duration	Ex. Ca	Ex. K	Ex. Mg	Ex. Na	Av. Cu	Av. Fe	Av. Mn	Av. Zn	
(a) 0–10 cm										
ANPP	2-year	-0.33	0.34	-0.41*	-0.22	0.02	-0.08	0.73***	-0.16	
	5-year	-0.73***	-0.27	-0.34	-0.41*	0.85***	0.80***	0.80***	0.20	
	10-year	-0.23	-0.36	-0.07	0.04	0.48*	0.14	0.24	0.16	
	all	-0.11	0.12	0.30**	0.22	0.01	0.18	0.27*	-0.35**	
BGB	2-year	0.01	0.01	0.22	0.01	0.05	-0.10	0.04	0.30	
	5-year	-0.49*	-0.37	-0.19	-0.13	0.46*	0.40*	0.45*	-0.13	
	10-year	-0.29	0.09	-0.28	-0.05	0.59**	0.35	0.42*	0.45*	
	all	-0.10	-0.17	-0.22	-0.02	0.10	0.12	0.10	0.05	
pН	2-year	0.39	-0.32	0.37	0.38	-0.06	0.05	-0.74***	0.01	
	5-year	0.85***	0.28	0.32	0.61**	-0.82***	-0.85***	-0.88***	-0.18	
	10-year	0.96***	0.16	0.86***	0.43*	-0.86	-0.95***	-0.96***	-0.77***	
	all	0.81***	0.11	0.49***	0.48***	-0.64***	-0.81***	-0.85***	-0.45***	
(b) 10–20 cm										
ANPP	2-year	-0.32	-0.06	0.07	-0.50*	-0.38	-0.41*	0.14	-0.20	
	5-year	-0.48*	0.30	0.59**	-0.66***	0.30	0.15	0.78***	-0.14	
	10-year	-0.38	-0.31	-0.09	-0.02	0.10	0.24	0.08	0.27	
	all	-0.26*	-0.35**	0.55***	-0.06	-0.52***	-0.64***	0.26*	-0.05	
BGB	2-year	0.20	-0.31	0.18	0.22	-0.11	-0.04	-0.06	-0.24	
	5-year	-0.02	0.19	0.33	-0.13	-0.14	-0.35	0.33	-0.08	
	10-year	-0.18	0.36	0.16	0.13	0.26	0.29	0.19	0.56**	
	all	0.21	-0.31**	0.25*	0.53***	-0.48***	-0.47***	-0.24*	-0.19	
pН	2-year	0.28	0.23	0.06	0.09	0.20	0.29	0.15	0.34	
	5-year	0.51*	-0.25	-0.60**	0.75***	-0.32	-0.22	-0.80***	0.15	
	10-year	0.88***	-0.19	0.45*	0.68***	-0.84***	-0.74***	-0.93***	0.34	
	all	0.63***	-0.40	0.36**	0.59***	-0.74***	-0.64***	-0.76***	0.16	

Note. See note in Table 1.

### 4.3. The Intensity and Duration of N Addition Jointly Affect Soil Micronutrients

Contrary to our second and third hypothesis, our results showed that the available contents of micronutrients increased only under relatively higher rates combined with longer duration of N addition while generally decreased under low- and mid-N rates and with short term N addition (Figure 3). These results might indicate that the relative importance of the effects of pH and plant uptake on these available micronutrients changed with prolonged N addition duration. Based on our results (Figure 4 and Table 2), effects of plant uptake of available micronutrients as indicated by significant increase in ANPP was stronger than that of soil pH at the early stage of N addition, while tended to be weaker as the ANPP was less sensitive to longer N addition durations (Table S1 in Supporting Information S1). Hence, under higher rate and long duration of N addition, the sharp increase of micronutrient contents was determined mainly by the increasing severity of soil acidification, which was also supported by our SEM analysis. Moreover, the notably high content of available Mn with high N addition and negative relationship between available Mn and soil pH after short-term N addition (Figure 2 and Table 2) demonstrated that response of available Mn was more sensitive to soil pH than those of Fe, Cu, and Zn (Alejandro et al., 2020; Andresen et al., 2018).



Although we lack the data of plant micronutrients, there are good evidences to show that plants are more likely to uptake more micronutrients (i.e., Mn) in leaves under chronic N addition especially in N-limited grassland ecosystems (Cai et al., 2017; Kao et al., 2020; Tian et al., 2020; Verbruggen & Hermans, 2013). Our analysis also suggested that soil pH is an important factor in affecting soil micronutrient availability, and its influence increased with increasing duration of N addition (Figures 4d-4f). This is because soil pH largely governs the sorption-desorption and oxidation-reduction processes of micronutrients, and further influences their availability (Kao et al., 2020). For instance, Fe and Mn generally tend to present as more soluble forms, that is, Fe/Mn oxides/hydroxide when soil pH decreases (Kao et al., 2020). However, their sorption with clay minerals and organic matter will become stronger when pH increases (Kao et al., 2020). In this study, long-term N addition significantly decreased soil pH by 0.84 unit (mean value across all N addition rates) in the 0-10 cm soil and by 0.48 unit in the 10–20 cm soil after long-term N addition (Table S1 in Supporting Information S1). Moreover, the deceasing effects on soil pH were magnified with increasing intensity and duration of N addition (Table S1 in Supporting Information S1). No obvious changes in micronutrients were found in low-N addition even under long-term N addition. This is possibly because the decreasing effects on these elements induced by plant uptake offset the increasing effects induced by changes in soil pH (Alejandro et al., 2020; Cai et al., 2017; Liu et al., 2021). The consistent increase of available Fe, Mn and Cu to mid- and long-term N addition reflect the increment of soil pH effects over time (Andresen et al., 2018; Fang et al., 2019), which is supported by our VPA and SEM analysis (Figure 4). B. Wang et al. (2017) also showed that the 9-year N addition in a field experiment increased the soil available Fe, Mn and Cu, and they were negatively correlated with soil pH. Overall, our results suggest that the relative influences of pH and plant growth on nutrients, especially on the micronutrients showed remarkable differences among different N addition durations.

### 4.4. Implications and Perspectives

Collectively, based on a long-term continuous field N addition experiment in temperate grasslands, we determined how intensity and duration of N addition individually and interactively affected the contents of mineral nutrients, and explored roles of plant uptake and soil pH. Our results indicated that micronutrients had quick responses, and the excessive accumulation of some specific elements such as Mn in plants may lead to lower nutritive value of forage and livestock as well (Kao et al., 2020; B. Wang et al., 2021). These results can also help to explain the inconsistent findings among meta-analysis studies and may be helpful in improving the predictability of biogeochemical models (Lucas et al., 2011; Meng et al., 2019). However, more studies are still needed to further verify these findings. For instance, it will be more convincing when the time period of study is long enough and include at least years with remarkably different size of precipitation (Ren et al., 2021). Additionally, we did not explore the difference in responses to N addition between the two soil layers (i.e., 0–10 cm vs. 10–20 cm) as their responses were generally similar, but deep soils are deserved more attention (Koarashi et al., 2012; Z. P. Yu et al., 2020). Finally, we only assessed the relative influence of pH and plant growth on the dynamics of mineral nutrients (Greaver et al., 2016; Horswill et al., 2008; Phoenix et al., 2020). Therefore, further studies should incorporate more explainable factors (such as plant nutrient status) into the statistical models and combine soil acidification experiment or liming experiments to explore the underlying mechanisms.

# 5. Conclusions

Our results demonstrated that N addition significantly altered the contents of 10 mineral nutrients in surface soils (0–20 cm) of a temperate grassland, and these effects varied with intensity and duration of N addition. Experiments with only short-term N addition could not accurately forecast the dynamics of mineral nutrients in grassland ecosystems since the decreasing effects of N addition on TP and base cations, or the increasing positive effects on TN and micronutrients generally occur under longer term N addition. The response of micronutrients may reflect that the effects of plant uptake on micronutrients were stronger than that of pH at the early stage of N addition, while the role of soil pH is increasingly important with prolonged duration of N addition. These results addressed the importance of accumulative N-addition effect and suggest that it is essential to consider effects of treatment duration when building soil biogeochemical models.



Acknowledgments

# **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

# **Data Availability Statement**

Associated data are listed in Supporting Information S1. And the data are also archived in Figshare (https://figshare.com/s/9b5e4210d36d16ff470a). The DOI becomes active when the item is published (DOI: 10.6084/m9.figshare.17351771).

# References

- Alejandro, S., Höller, S., Meier, B., & Peiter, E. (2020). Manganese in plants: From acquisition to subcellular allocation. Frontiers in Plant Science, 11, 00300. https://doi.org/10.3389/fpls.2020.00300
- Andresen, E., Peiter, E., & Küpper, H. (2018). Trace metal metabolism in plants. Journal of Experimental Botany, 69, 909–954. https://doi.org/10.1093/jxb/erx465
- Bai, Y. F., Wu, J. G., Clark, C. M., Naeem, S., Pan, Q. M., Huang, J. H., et al. (2010). Tradeoffs and thresholds in the effects of nitrogen addition on biodiversity and ecosystem functioning: Evidence from inner Mongolia grasslands. *Global Change Biology*, 16, 358–372. https://doi. org/10.1111/j.1365-2486.2009.01950.x
- Barrow, N. J. (2017). The effects of pH on phosphate uptake from the soil. *Plant and Soil*, 410, 401–410. https://doi.org/10.1007/s11104-016-3008-9
  Bengtsson, J., Bullock, J. M., Egoh, B., Everson, C., Everson, T., O'Connor, T., et al. (2019). Grasslands—More important for ecosystem services than you might think. *Ecosphere*, 10, e02582. https://doi.org/10.1002/ecs2.2582
- Bowman, W. D., Cleveland, C. C., Halada, L., Hresko, J., & Baron, J. S. (2008). Negative impact of nitrogen deposition on soil buffering capacity. *Nature Geoscience*, 1, 767–770. https://doi.org/10.1038/ngeo339

Boy-Roura, M., Cameron, K. C., & Di, H. J. (2015). Identification of nitrate leaching loss indicators through regression methods based on a meta-analysis of lysimeter studies. *Environmental Science and Pollution Research*, 23, 3671–3680. https://doi.org/10.1007/s11356-015-5529-9

- Bünemann, E. K., Bongiorno, G., Bai, Z. G., Creamer, R. E., De Deyn, G., deGoede, R., & Brussaard, L. (2018). Soil quality—A critical review. Soil Biology and Biochemistry, 120, 105–125.
- Cai, J. P., Weiner, J., Wang, R. Z., Luo, W. T., Zhang, Y. Y., Liu, H. Y., & Jiang, Y. (2017). Effects of nitrogen and water addition on trace element stoichiometry in five grassland species. *Journal of Plant Research*, 130, 1–10. https://doi.org/10.1007/s10265-017-0928-2
- Cameron, K. C., Di, H. J., & Moir, J. L. (2013). Nitrogen losses from the soil/plant system: A review. Annals of Applied Biology, 162, 145–173. https://doi.org/10.1111/aab.12014
- Chen, J., Groenigen, K. J., Hungate, B. A., Terrer, C., Groenigen, J. W., Maestre, F. T., et al. (2020). Long-term nitrogen loading alleviates phosphorus limitation in terrestrial ecosystems. *Global Change Biology*, 26, 5077–5086. https://doi.org/10.1111/gcb.15218
- Clark, C. M., & Tilman, D. (2008). Loss of plant species after chronic low-level nitrogen deposition to prairie grasslands. *Nature*, 405, 06503. https://doi.org/10.1038/nature06503
- Deng, Q., Hui, D. F., Dennis, A., Reddy, K. C., & Xu, X. (2017). Responses of terrestrial ecosystem phosphorus cycling to nitrogen addition: A meta-analysis. *Global Ecology and Biogeography*, 26, 713–728. https://doi.org/10.1111/geb.12576
- de Oliveira Garcia, W., Amann, T., Hartmann, J., Karstens, K., Popp, A., Boysen, L. R., et al. (2020). Impacts of enhanced weathering on biomass production for negative emission technologies and soil hydrology. *Biogeosciences*, 17, 2107–2133. https://doi.org/10.5194/bg-17-2107-2020
- Doetterl, S., Berhe, A. A., Arnold, C., Bodé, S., Fiener, P., Finke, P., et al. (2018). Links among warming, carbon and microbial dynamics mediated by soil mineral weathering. *Nature Geoscience*, 11, 589–593. https://doi.org/10.1038/s41561-018-0168-7
- Du, E. Z., Terrer, C., Pellegrini, A. F. A., Ahlström, A., vanLissa, C. J., Zhao, X., et al. (2020). Global patterns of terrestrial nitrogen and phosphorus limitation. *Nature Geoscience*, 13, 221–226. https://doi.org/10.1038/s41561-019-0530-4
- Fang, K., Kou, D., Wang, G. Q., Chen, L. Y., Ding, J. Z., Li, F., et al. (2017). Decreased soil cation exchange capacity across northern China's grasslands over the last three decades. *Journal of Geophysical Research: Biogeosciences*, 122, 3088–3097. https://doi.org/10.1002/2017jg003968
- Fang, K., Qin, S., Chen, L., Zhang, Q., & Yang, Y. (2019). Al/Fe mineral controls on soil organic carbon stock across Tibetan alpine grasslands. Journal of Geophysical Research: Biogeosciences, 124, 247–259. https://doi.org/10.1029/2018jg004782
- Fowler, D., Coyle, M., Skiba, U., Sutton, M. A., Cape, J. N., Reis, S., et al. (2013). The global nitrogen cycle in the twenty-first century. *Philosophocal Transactions of the Royal Society B-Biological Sciences*, 368, 20130164. https://doi.org/10.1098/rstb.2013.0164
- Galloway, J., Townsend, A., Erisman, J., Bekunda, M., Cai, Z., Freney, J., et al. (2008). Transformation of the nitrogen cycle: Recent trends, questions, and potential solutions. *Science*, 320, 889–892. https://doi.org/10.1126/science.1136674
- Greaver, T. L., Clark, C. M., Compton, J. E., Vallano, D., Talhelm, A. F., Weaver, C. P., et al. (2016). Key ecological responses to nitrogen are altered by climate change. *Nature Climate Change*, 9, 836–843. https://doi.org/10.1038/nclimate3088
- Haddad, K. S., & Evans, J. C. (1993). Assessment of chemical methods for extracting zinc, manganese, copper, and iron from New South Wales soils. *Communications in Soil Science and Plant Analysis*, 24, 29–44. https://doi.org/10.1080/00103629309368779
- Horswill, P., O'Sullivan, O., Phoenix, G. K., Lee, J. A., & Leake, J. R. (2008). Base cation depletion, eutrophication and acidification of speciesrich grasslands in response to long-term simulated nitrogen deposition. *Environmental Pollution*, 155, 336–349. https://doi.org/10.1016/j. envpol.2007.11.006
- Hou, E. Q., Chen, C. R., Luo, Y. Q., Zhou, G. Y., Kuang, Y. W., Zhang, Y. G., et al. (2018). Effects of climate on soil phosphorus cycle and availability in natural terrestrial ecosystems. *Global Change Biology*, 24, 3344–3356. https://doi.org/10.1111/gcb.14093
- Hou, E. Q., Wen, D. L., Jiang, L. F., Luo, X. Z., Kuang, Y. W., Lu, X. K., et al. (2021). Latitudinal patterns of terrestrial phosphorus limitation over the globe. *Ecology Letters*, 24, 1420–1431. https://doi.org/10.1111/ele.13761
- Kao, P. T., Darch, T., McGrathc, S. P., Kendall, N. R., Buss, H. L., Warrene, H., & Lee, M. R. F. (2020). Book: Chapter 4, Factors influencing elemental micronutrient supply from pasture systems for grazing ruminants. Advances in Agronomy, 164, 161–229. https://doi.org/10.1016/ bs.agron.2020.06.004
- Keiblinger, K. M., Schneider, M., Gorfer, M., Paumann, M., Deltedesco, E., Berger, H., et al. (2018). Assessment of Cu applications in two contrasting soils-effects on soil microbial activity and the fungal community structure. *Ecotoxicol*, 27, 217–233. https://doi.org/10.1007/s10646-017-1888-y

### We thank Ms. Canran Yang, Yiqian Jin, Buqing Zhong, and Zengjuan Fu for their assistance in data analyses and laboratory analysis. We also thank the subject editor and the two anonymous reviewers to give us the constructive comments on our manuscript. QSN was supported by the Open Foundation of the State Key Laboratory of Urban and Regional Ecology of China, and the open Foundation of the State Key Laboratory of Grassland Agro-ecosystems of China. This work was financially supported by the National Natural Science Foundation of China (31870440, 32071562).

21698961

- Koarashi, J., Hockaday, W. C., Masiello, C. A., & Trumbore, S. E. (2012). Dynamics of decadally cycling carbon in subsurface soils. Journal of Geophysical Research, 117, G03033. https://doi.org/10.1029/2012jg002034
- Li, Y., Niu, S. L., & Yu, G. R. (2016). Aggravated phosphorus limitation on biomass production under increasing nitrogen loading: A meta-analysis. Global Change Biology, 22, 934–943. https://doi.org/10.1111/gcb.13125
- Lindsay, W. L., & Norvell, W. A. (1978). Development of a DTPA soil test for zinc, iron, manganese, and copper. Soil Science Society of America Journal, 4, 421–428. https://doi.org/10.2136/sssaj1978.03615995004200030009x
- Liu, H. Y., Wang, R. Z., Lü, X. T., Cai, J. P., Li, H., Zhang, Y. G., et al. (2021). Effects of nitrogen addition on plant-soil micronutrients vary with nitrogen form and mowing management in a meadow steppe. *Environmental Pollution*, 289, 117969. https://doi.org/10.1016/j. envpol.2021.117969
- Lu, M., Yang, Y., Luo, Y., Fang, C., Zhou, X., Chen, J., et al. (2011). Responses of ecosystem nitrogen cycle to nitrogen addition: A meta-analysis. New Phytologist, 189, 1040–1050. https://doi.org/10.1111/j.1469-8137.2010.03563.x
- Lu, X. K., Vitousek, P. M., Mao, Q. G., Gilliam, F. S., Luo, Y. Q., Zhou, G. Y., et al. (2018). Plant acclimation to long-term high nitrogen deposition in an N-rich tropical forest. Proceedings of National Academy of Sciences of the United States of America, 115, 5187–5192. https://doi. org/10.1073/pnas.1720777115
- Lucas, R. W., Klaminder, J., Futter, M. N., Bishop, K. H., Egnell, G., Laudon, H., & Hogberg, P. (2011). A meta-analysis of the effects of nitrogen additions on base cations: Implications for plants, soils, and streams. *Forest Ecology and Management*, 262, 95–104. https://doi.org/10.1016/j. foreco.2011.03.018
- Marklein, A. R., & Houlton, B. Z. (2012). Nitrogen inputs accelerate phosphorus cycling rates across a wide variety of terrestrial ecosystems. New Phytologost, 193, 697–704. https://doi.org/10.1111/j.1469-8137.2011.03967.x
- Marschner, H. (2012). Marschner's mineral nutrition of higher plants (Vol. 193, pp. 696-704). Academic press, New Phytologost.
- Meng, C., Tian, D. S., Zeng, H., Li, Z. L., Yi, C. X., & Niu, S. L. (2019). Global soil acidification impacts on belowground processes. *Environmental Research Letters*, 14, 074003. https://doi.org/10.1088/1748-9326/ab239c
- Niu, D. C., Yuan, X. B., Cease, A. J., Wen, H. Y., Zhang, C. P., Fu, H., & Elser, J. J. (2018). The impact of nitrogen enrichment on grassland ecosystem stability depends on nitrogen addition level. *Science of the Total Environment*, 618, 1529–1538. https://doi.org/10.1016/j. scitotenv.2017.09.318
- Niu, G. X., Hasi, M. Q. E., Wang, R. Z., Wang, Y. L., Geng, Q. Q., Hu, S. Y., et al. (2021). Soil microbial community responses to long-term nitrogen addition at different soil depths in a typical steppe. *Applied Soil Ecology*, 167, 104054. https://doi.org/10.1016/j.apsoil.2021.104054
- Niu, G. X., Wang, R. Z., Hasi, M. Q. E., Wang, Y. L., Geng, Q. Q., Wang, C. H., et al. (2021). Availability of soil base cations and micronutrients along soil profile after 13-year nitrogen and water addition in a semi-arid grassland. *Biogeochemistry*, 152, 223–236. https://doi.org/10.1007/ s10533-020-00749-5
- Ochoa-Hueso, R., Bell, M. D., & Manrique, E. (2014). Impacts of increased nitrogen deposition and altered precipitation regimes on soil fertility and functioning in semiarid Mediterranean shrublands. *Journal of Arid Environments*, 104, 106–115. https://doi.org/10.1016/j. jaridenv.2014.01.020
- Phoenix, G. K., Emmett, B. A., Britton, A. J., Caporn, S. A., Dise, N. B., Helliwell, R., et al. (2012). Impacts of atmospheric nitrogen deposition: Responses of multiple plant and soil parameters across contrasting ecosystems in long-term field experiments. *Global Change Biology*, 18, 1197–1215. https://doi.org/10.1111/j.1365-2486.2011.02590.x
- Phoenix, G. K., Johnson, D. A., Muddimer, S. P., Leake, J. R., & Cameron, D. D. (2020). Niche differentiation and plasticity in soil phosphorus acquisition among co-occurring plants. *Nature Plants*, 6, 349–354. https://doi.org/10.1038/s41477-020-0624-4
- Rafiei, M., & Kennedy, M. (2020). Weathering in a world without terrestrial life recorded in the Mesoproterozoic Velkerri Formation. *Nature Communications*, 10, 3448.
- Ren, Z. R., Zhang, Y. Q., & Zhang, Y. H. (2021). Nitrogen deposition magnifies the positive response of plant community production to precipitation: Ammonium to nitrate ratio matters. *Environmental Pollution*, 276, 116659. https://doi.org/10.1016/j.envpol.2021.116659
- Spohn, M. (2020). Increasing the organic carbon stocks in mineral soils sequesters large amounts of phosphorus. *Global Change Biology*, 26, 4169–4177. https://doi.org/10.1111/gcb.15154
- Stevens, C. J. (2015). Anthropogenic nitrogen deposition predicts local grassland primary production worldwide. *Ecology*, 96, 1495–1465. https://doi.org/10.1890/14-1902.1
- Tian, D. S., Reich, P. B., Chen, H. Y. H., Xiang, Y. Z., Luo, Y. Q., Shen, Y., et al. (2019). Global changes alter plant multi-element stoichiometric coupling. New Phytologost, 221, 807–817. https://doi.org/10.1111/nph.15428
- Tian, Q. Y., Yang, L. Y., Ma, P. F., Zhou, H. R., Liu, N. N., Bai, W. M., et al. (2020). Below-ground-mediated and phase-dependent processes drive nitrogen-evoked community changes in grasslands. *Journal of Ecology*, 108, 1874–1887. https://doi.org/10.1111/1365-2745.13415
- Verbruggen, N., & Hermans, C. (2013). Physiological and molecular responses to magnesium nutritional imbalance in plants. *Plant and Soil, 368*, 87–99. https://doi.org/10.1007/s11104-013-1589-0
- Vitousek, P. M. (2015). Grassland ecology: Complexity of nutrient constraints. *Nature Plants*, *1*, 15098. https://doi.org/10.1038/nplants.2015.98
   Vitousek, P. M., Naylor, R., Crews, T., David, M. B., Drinkwater, L. E., Holland, E., et al. (2009). Nutrient imbalances in agricultural development. *Science*, *324*, 1519–1520. https://doi.org/10.1126/science.1170261
- Waldron, K. J., Rutherford, J. C., Ford, D., & Robinson, N. J. (2009). Metalloproteins and metal sensing. *Nature*, 460, 823–830. https://doi. org/10.1038/nature08300
- Wang, B., Wu, Y., Chen, D., Hu, S. J., & Bai, Y. F. (2021). Legacy effect of grazing intensity mediates the bottom-up controls of resource addition on soil food webs. *Journal of Applied Ecology*, 58, 976–987. https://doi.org/10.1111/1365-2664.13825
- Wang, R., Dungait, J., Buss, H., Yang, S., Zhang, Y., Xu, Z., & Jiang, Y. (2017). Base cations and micronutrients in soil aggregates as affected by enhanced nitrogen and water inputs in a semi-arid steppe grassland. *Science of the Total Environment*, 575, 564–572. https://doi.org/10.1016/j. scitotenv.2016.09.018
- Wang, R., Zhang, Y., He, P., Yin, J., Yang, J., Liu, H., et al. (2018). Intensity and frequency of nitrogen addition alter soil chemical properties depending on mowing management in a temperate steppe. *Journal of Environmental Management*, 224, 77–86. https://doi.org/10.1016/j. jenvman.2018.07.036
- Watanabe, T., Broadley, M. R., Jansen, S., White, P. J., Takada, J., Satake, K., et al. (2007). Evolutionary control of leaf element composition in plants. New Phytologost, 174, 516–523. https://doi.org/10.1111/j.1469-8137.2007.02078.x
- Wen, Z., Xu, W., Li, Q., Han, M. J., Tang, A. H., Zhang, Y., et al. (2020). Changes of nitrogen deposition in China from 1980 to 2018. Environment International, 144, 106022. https://doi.org/10.1016/j.envint.2020.106022
- Yu, G. R., Jia, Y. L., He, N. P., Zhu, J. X., Chen, Z., Wang, Q. F., et al. (2019). Stabilization of atmospheric nitrogen deposition in China over the past decade. *Nature Geoscience*, 12, 424–429. https://doi.org/10.1038/s41561-019-0352-4



- Yu, Z. P., Chen Han, Y. H., Seare, E. B., Sardans, J., Ciais, P., Peñuelas, J., & Huang, Z. Q. (2020). Whole soil acidification and base cation reduction across subtropical China. *Geoderma*, 361, 114107. https://doi.org/10.1016/j.geoderma.2019.114107
- Zhang, J., Cai, Z., & Müller, C. (2018). Terrestrial N cycling associated with climate and plant-speciic N preferences: A review. European Journal of Soil Science, 69, 488–501. https://doi.org/10.1111/ejss.12533
- Zia, A., vanden berg, L., Riaz, M., Arif, M., Zia, D., Khan, A. J., et al. (2020). Nitrogen induced DOC and heavy metals leaching: Effects of nitrogen forms, deposition loads and liming. *Environmental Pollution*, 265, 114981. https://doi.org/10.1016/j.envpol.2020.114981